

# OPEN ACCESS

**Citation:** Dearlove BL, Frost SDW (2015) Measuring Asymmetry in Time-Stamped Phylogenies. PLoS Comput Biol 11(7): e1004312. doi:10.1371/journal. pcbi.1004312

**Editor:** Sergei L. Kosakovsky Pond, University of California San Diego, UNITED STATES

Received: December 9, 2014

Accepted: May 4, 2015

Published: July 6, 2015

**Copyright:** © 2015 Dearlove, Frost. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The observed HIV, influenza and ebola trees (in Newick format), treelmbalance R package, and example R script to create Figs <u>4</u>, <u>5</u> and <u>6</u> are all available from the Dryad Digital Repository: <u>http://dx.doi.org/10.5061/dryad.</u> <u>v7817</u>.

**Funding:** This work was supported by a Medical Research Council Methodology Research Programme grant to SDWF (grant number MR/ J013862/1). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE** 

# Measuring Asymmetry in Time-Stamped Phylogenies

#### Bethany L. Dearlove\*, Simon D. W. Frost

Department of Veterinary Medicine, University of Cambridge, Cambridge, United Kingdom

\* bd357@cam.ac.uk

# Abstract

Previous work has shown that asymmetry in viral phylogenies may be indicative of heterogeneity in transmission, for example due to acute HIV infection or the presence of 'core groups' with higher contact rates. Hence, evidence of asymmetry may provide clues to underlying population structure, even when direct information on, for example, stage of infection or contact rates, are missing. However, current tests of phylogenetic asymmetry (a) suffer from false positives when the tips of the phylogeny are sampled at different times and (b) only test for global asymmetry, and hence suffer from false negatives when asymmetry is localised to part of a phylogeny. We present a simple permutation-based approach for testing for asymmetry in a phylogeny, where we compare the observed phylogeny with random phylogenies with the same sampling and coalescence times, to reduce the false positive rate. We also demonstrate how profiles of measures of asymmetry calculated over a range of evolutionary times in the phylogeny can be used to identify local asymmetry. In combination with different metrics of asymmetry, this combined approach offers detailed insights of how phylogenies reconstructed from real viral datasets may deviate from the simplistic assumptions of commonly used coalescent and birth-death process models.

# Author Summary

Phylogenetic trees of viruses sampled from different individuals provide clues to the dynamics of transmission. The extent to which the tree is asymmetric may be influenced by biological factors such as differences in infectiousness or contact rates between individuals, but also by nuisance factors such as the pattern of sampling. We have devised a simple statistical test for asymmetry, which controls for sampling patterns and potentially complex temporal dynamics by conditioning on the sampling and coalescence times in a phylogeny, and can also detect whether specific clades in the phylogeny drive patterns of asymmetry. We apply our approach to data on HIV, influenza A virus H5N1, and ebola virus. **Competing Interests:** The authors have declared that no competing interests exist.

#### Introduction

Genetic approaches to investigating infectious diseases are well-established, exploiting the naturally high genetic diversity in pathogen populations such as HIV and influenza to reconstruct both their evolutionary and epidemiological dynamics [1]. Phylogenies contain potentially large amounts of information on disease dynamics, and can help reveal the disease incidence and prevalence, changes in historical population size, and population substructure [2–4]. However, there can be confounding factors when trying to convert evolutionary dynamics into epidemiological quantities such as transmission rates, and ideally we want to be able to explicitly model viral transmission in an evolutionary framework, taking into account features such as the host population structure (for example, differences in contact rates between groups of individuals) and the natural course of infection (for example, differences in infectiousness during the acute and chronic phases of HIV infection) [5].

One way to investigate the extra biological complexity of such patterns is to consider the shape or branching structure of the phylogeny, a feature that is arguably underused despite being relatively straightforward to infer. Evidence of asymmetry in a tree reflects heterogeneity in the population that has arisen due to the processes by which a tree has grown [6]; previous work suggests that evidence of asymmetry in a phylogenetic tree can arise due to selection [2], heterogeneity in contact rates [7] and population structure [5]. Since many tree models assume homogeneity in the population, it is important to be able to identify which parts of the tree might be driving asymmetry, and whether or not this is problematic under the modelling assumptions—preferably before running computationally expensive analyses.

It is common to analyse viral datasets sampled over multiple timepoints. As viruses, including RNA and ssDNA viruses, evolve rapidly, phylogenetic reconstruction gives rise to trees with root-to-tip distances that reflect, in part, sampling times. However, such trees are more likely to be asymmetric, resulting in standard metrics developed for homochronous sampling being implicitly biased (see Supplementary Information of Frost and Volz (2013) [5]). This is due to the fact that most metrics use the topological distance (that is, the number of nodes traversed between two points in the tree), and isolates sampled earlier in the history of the phylogeny will tend to have fewer nodes between them and the root of the tree.

In this paper, we propose a permutation-based approach that allows an observed phylogeny to be compared to random phylogenies with the same sampling and coalescence times. This approach can also be used to assess asymmetry throughout evolutionary history in a rooted tree, therefore also allowing areas of local asymmetry to be identified in addition to a single global value at the root of the tree. We demonstrate this approach on three datasets with different expected types of heterogeneity, illustrating the imprint of various transmission dynamics on viral phylogenies.

# **Materials and Methods**

## Measuring phylogenetic tree shape

There are a number of ways to measure the balance of a phylogeny. Most approaches consider either the topological distance (the number of nodes) between two parts of the tree, for example Sackin's index, or the balance of each internal node by comparing the number of leaves in the left and right subtrees below it, for example Colless' index [8-11]. Here, we consider two measures of asymmetry: Sackin's index [11], and the number of cherries [12], although the methodology can easily be extended to other metrics.

Sackin's index is the total topological distance between the leaves and root of the tree. If  $d_j$  is the number of nodes to be traversed between each leaf j and the root, then Sackin's index is the

total over all leaves,

$$I_{\rm S} = \sum_j d_j. \tag{1}$$

In trees where the tips have been sampled at the same time, the expected Sackin's index,  $\mathbb{E}(I_S(n))$ , for *n* isolates in the sample is given by:

$$\mathbb{E}(I_{s}(n)) = 2n \sum_{k=2}^{n} \frac{1}{k}$$
<sup>(2)</sup>

under the Yule or coalescent models [13]. For large n,  $\mathbb{E}(I_S(n)) \approx 2n \log(n)$ . Since the expected value of the Sackin's index increases with the tree sample size, it is common to either divide the statistic by n (i.e. the mean topological distance from root to tip), or use the following standar-disation proposed by Leventhal et al. [7]:

$$\overline{I}_{s}(n) = \frac{I_{s}(n) - \mathbb{E}(I_{s}(n))}{\mathbb{E}(I_{s}(n))}.$$
(3)

However, since the permutation method outlined in this paper compares an observed tree to those of the same size (i.e. like with like), we simply use the non-standardised version here. We use the function sackin.test in the apTreeshape R package to test the hypothesis of asymmetry in the tree, comparing the observed value to 10,000 trees simulated under the Yule model [14, 15].

A cherry is formed when two tips share a direct ancestor. In an asymmetric tree, tips generally coalesce with branches earlier in the ancestry of the tree, and therefore fewer cherries are expected than with a balanced tree. Under the Yule or coalescent model, the expected number of cherries,  $C_n$ , in a tree with *n* taxa is *n*/3, and for a uniform tree is *n*/4 [12]. In addition, McKenzie and Steel showed that the number of cherries is asymptotically normal with

$$\frac{C_n - n/3}{\sqrt{2n/45}} \to \mathcal{N}(0, 1) \tag{4}$$

under the Yule or coalescent model, and

$$\frac{C_n - n/4}{\sqrt{n/16}} \to \mathcal{N}(0, 1) \tag{5}$$

for a uniform tree [12].

These two metrics complement each other well, as the number of cherries reveals recent asymmetry in the tree, whereas Sackin's index gives the asymmetry of the tree over the whole evolutionary history [5]. In addition, these metrics are only weakly correlated, unlike for example, the Sackin and Colless indices [16, 17].

#### Local asymmetry

The ordering of nodes in a rooted tree means we can consider the asymmetry in the phylogenetic tree throughout the evolutionary period, and not just at the root. This asymmetry could be due to a small effect at each internal node accumulating throughout the tree, or due to one or more nodes with highly imbalanced subtrees below them. Calculating the asymmetry over the entire course of the tree allows us to identify local asymmetry, even when there may not be significant evidence for global asymmetry (as obtained by considering the cumulative statistics at the root of the tree). There are two main types of event that can affect the shape of a phylogeny: a coalescence, and a new sampling event, which adds a tip. Sackin's index and the number of cherries are both concerned with internal nodes rather than the tips, so we need only consider the former. At each coalescent event, we consider the contribution of that node to the overall metric. This results in a vector of n - 1 values, one for each ancestral node, giving a measure of how asymmetric the subtree below the node is (Fig 1). We can add these values cumulatively as we go backwards in time from the present towards the root, to investigate how asymmetry builds up over the course of the tree.

For the number of cherries, calculating the effect of each individual node is straightforward —being 1 if the node is a cherry (i.e. the direct ancestor of two tips) and 0 if it is not. To calculate the Sackin's index for each node, rather than count the topological distance to the root for each tip as the calculation of the Sackin's index is usually presented, we instead consider the number of times each node is traversed going from the tip to the root. Namely, this is the number of tips found in the subtree below the node of interest.

## Permuting the tree

To obtain the distribution of possible values for each the statistics for an observed tree, we permute the tree whilst retaining the same tip sampling and internal node times ( $\underline{Fig.2}$ ). These simulated trees form a neutrally evolving null distribution of coalescent trees, conditioned on the same tip and internal node times as the observed tree.

For *n* tips, there are n - 1 internal nodes. Starting at the time of the most recent internal node (say,  $t_1$ ) and going backwards in time from the present at t = 0, we consider all tips that were sampled more recently (i.e. between  $t_1$  and t = 0). Two of these tips are then chosen at random to coalesce, thus creating the internal node for  $t_1$ . This continues backwards in time for each node in turn, with the only difference that coalescences can be between sample tips and nodes that have already been produced via a coalescence between the time to node *i*,  $t_i$ , and the present.

The code to simulate permutations of an observed tree with the same sampling and coalescence times, and all the imbalance metrics considered above, were written in R [15], and are available as part of the treeImbalance package on GitHub (<u>https://github.com/bdearlove/</u> <u>treeImbalance</u>), and are in the Dryad Digital Repository: <u>http://dx.doi.org/10.5061/dryad.</u> <u>v7817</u>.

## Hypothesis testing

To obtain a distribution of possible values for each imbalance metric, 10,000 permutations of the observed tree with the same tip sampling and internal node times were generated. For each of these permuted trees, the number of cherries and Sackin's index were calculated at each internal node and globally by computing the cumulative statistics at the root (Fig 1).

The median trajectory of Sackin's index and the number of cherries throughout the ancestry of the tree (shown with a solid red line in plots) was calculated by partitioning around the medoid with a single cluster using the function pam in the cluster R package [18]. The medoid represents the trajectory which has the least dissimilarity with all the other possible trajectories from the permutation test. This ensures that the median is obtained from within the set of permuted trajectories, thus ensuring it is a 'viable' trajectory, and overcomes issues associated with other methods (such as calculating the mean or median statistic at each node), which do not necessarily force the trajectory to be monotonically increasing.

At each internal node, the 95% confidence interval was calculated by inverting the hypothesis test around the medoid value at that timepoint [19]. The medoid was subtracted from the









permuted trees, and then the critical points of this distribution are found where 2.5% of the values are as or more extreme (with no interpolation). The confidence interval then is obtained by adding these back to the medoid. Calculating the 95% confidence in this way, as opposed to using quantiles or the variance, ensures that the value calculated is within the permuted dataset. Since considering the local imbalance at each node results in a multiple hypothesis test, several p-value adjustments were considered in order to control the family-wise error rate (including the Bonferroni correction and methods proposed by Holm (1979), Hochberg (1988) and Hommel (1988) [20–22]), and the false discovery rate (including methods proposed by Benjamini and Hochberg (1995) and Benjamini and Yekutieli (2001) [23, 24]). For the latter, we also investigated the q-value, which estimates the proportion of significant hypotheses that are false [25–27]. Results were generally consistent (S1 Table), so here we report the most conservative adjustment, the Bonferroni correction, alongside the unadjusted p-values. The uncorrected values remain valuable since the purpose of the test is to identify potential deviations from the model for further investigation, rather than necessarily a strict hypothesis test.

For the cumulative statistics, the Bonferroni correction is equal to the number of internal nodes, n - 1. For the single node contribution to Sackin's index, the correction is n - 2, since at the root n tips will always be added.

# Simulated trees

To illustrate the bias of standard metrics, we simulated two sets of trees—one set with homochronous sampling (tips sampled at the same time) and one set with heterochronous sampling (tips sampled at different times). These were generated using Serial SimCoal [28] under a coalescent model with effective population size of  $10^4$ , with 100 tips sampled in the present for the homochronous sampling, and sampled over 10 time points each 1000 generations apart for the heterochronous sampling.

# **Phylogenies**

A single tip-dated phylogeny is required as input for our permutation approach. These can be obtained via a number of methods, but for viral datasets, the use of BEAST [29] is most common. Before implementing the permutation test, the observed trees were checked for polytomies, which were subsequently resolved into randomly ordered dichotomies with zero branch lengths. Negative branches were set equal to zero.

Tree files were available for the ebola virus [30] and influenza A virus [31] datasets in Newick format, and are available in the Dryad Digital Repository: <u>http://dx.doi.org/10.5061/dryad.</u> <u>v7817</u>. For the within-host HIV dataset [32], the sequences were aligned using MUSCLE v3.8.31 [33] and the maximum clade credibility tree (MCC) obtained using BEAST 1.8 with a GMRF Bayesian Skyride coalescent model [29]. The GTR model of nucleotide substitution [34] was used with an uncorrelated log-normal relaxed clock and a discretised gamma distribution with four categories was used to model rate heterogeneity across the sequence [35]. For the log-normal relaxed clock parameters, a uniform prior between 0.0 and  $1.0 \times 10^{100}$  was assumed for the mean, and an exponential with mean 1/3 for the standard deviation. A uniform (Dirichlet) prior was used for the nucleotide frequencies. The MCMC was run for 1 billion iterations, with a 10% burn-in period and samples saved every 10,000 iterations.

The within-host HIV skyride plot was obtained from the observed tree in R using an approximate approach that employs an integrated nested Laplace approximation [<u>36</u>].

# Results

In this section, we apply our test for detecting asymmetry in phylogenies with tips sampled at different times. The permutation test simulates new coalescent trees, conditional on the internal node and sampling times in the observed tree, so that a null distribution of asymmetry statistics can be calculated. The observed Sackin's index and number of cherries can then be compared with this distribution to assess how asymmetric the observed tree is, compared to trees with those times.

As an example, consider a tree simulated with 100 tips sampled over 10 time points (Fig 3a). Comparing this heterochronous tree with 1000 similarly simulated trees but with tips sampled at a single time point (Fig 3b) illustrates how extreme the observed values of Sackin's index and number of cherries (solid black line) are compared to the expected values (dashed black line), purely due to the serial sampling [5]. However, when the heterochronous observed tree is compared to a distribution obtained from the permuted trees (Fig 3c), it can be seen that in the distribution of possible trees with the same internal node and tip sampling times, there is little evidence to suggest that this observed tree is asymmetric.

We tested this pattern for a total of 100 similarly simulated trees with heterochronous tip sampling. Using the standard metrics, 99 trees were found to be more asymmetric than expected using Sackin's index, and 74 using the number of cherries. In contrast, using the permutation test with 10,000 simulated trees to control for the temporal signal, only two trees were





**Fig 3. Permutations of an observed tree can overcome bias in detecting asymmetry in time-sampled phylogenies.** a) An 'observed' tree, simulated under the coalescent model with 100 sequences sampled over 10 time points, each 1000 generations apart, with effective population size of 10<sup>4</sup>. b) The distribution of Sackin's index and number of cherries for 100 random trees, simulated as in a) except for tips being sampled at a single time point. Expected values for these distributions are shown with dashed black lines. The observed values (solid black line) are highly extreme due to the implicit bias caused by tips sampled early in the ancestry. However, this is not the case when comparing them to a distribution calculated from permuting the observed tree, as seen in c), where there is no evidence to suggest the observed tree is asymmetric and the solid black line falls between the 2.5% and 97.5% quantiles (dashed red lines).

significantly more asymmetric than expected with Sackin's index when compared to the 2.5% and 97.5% quantiles, and only one tree using the number of cherries.

#### Influenza A H5N1

We considered 98 influenza A virus H5N1 haemagglutinin sequences sampled from various bird species around seven locations in Asia (as distributed with BEAST v1.8.0, data originally collated by Wallace et al. (2007) [29, 31]). Here, we would reasonably expect that there could be three main sources of asymmetry in the phylogeny: the temporal sampling, selection and population substructure in the form of host species and location. Using the standard Sackin's index, the phylogeny is found to be extremely asymmetric (p-value <0.0001), though there is





Fig 4. Asymmetry in influenza A H5N1. a) Tree of 98 influenza A H5N1 haemagglutinin sequences sampled from bird species in Asia. b) Observed cumulative number of cherries over time (black), with results from permuted trees (grey). Inset histogram shows global results. Red lines show the medoid (solid) and 95% confidence interval of the permuted results (dashed). c) Trajectories for the cumulative Sackin's index. d) Node effect on Sackin's index over time. Nodes which are significant at an unadjusted p-value of 5% are shown by an open red circle.

not enough evidence to reject the null hypothesis of asymmetry at the tips using the number of cherries (p-value = 0.100).

However, when we condition on the heterochronous sampling and coalescence times using the permutation test, we find that there is no evidence for global asymmetry with either statistic (Fig 4). There may be evidence of individual nodes being more asymmetric than expected, with 12 nodes significant at an unadjusted significance level of 5%, though none remain significant after the Bonferroni correction (Fig 4a). This suggests that the extreme result seen with the standard Sackin's index was due to non-epidemiological effects, rather than heterogeneity in

the population. However, the unadjusted p-values may still hint towards a deviation from the model so it could be worth investigating a model that allows for heterogeneity.

## Ebola virus

The 2014 West Africa epidemic of ebola virus is the largest known outbreak of the virus, causing 25,791 cases and 10,689 deaths (as of 15th April 2015) across Guinea, Liberia and Sierra Leone [37]. A recent study by Gire et al. [30] investigated sequences from 78 patients in Sierra Leone, suggesting a central African source to the outbreak in 2004 with continued human-tohuman transmission, as opposed to punctuated re-transmission from a zoonotic source. A subsequent paper by Volz and Kosakovsky Pond [38] found strong evidence for superspreading, with much variance in the number of onward transmissions per individual, in contrast to the results of Stadler et al. who found that using two classes of transmission rates did not offer a significant improvement over an unstructured model [39]. Volz and Kosakovsky Pond note that this heterogeneity in transmission causes highly imbalanced phylogenies. Using a different method, Łuksza, Bedford and Lässig, identified a clade with a significantly higher growth rate than the ancestral clade it diverged from—again providing evidence for deviation from a simple randomly mixing model [40].

Similarly to the influenza data, the standard Sackin's index showed evidence of global asymmetry (p-value <0.0001), whilst the null hypothesis could not be rejected for the number of cherries (p-value = 0.141). Fig 5 shows the trajectory plots for the same statistics using the permutation test, showing that when controlling for the tip sampling being heterochronous, there is no evidence for asymmetry. Again, this suggests that the extreme result was due to non-epidemiological effects rather than heterogeneity in the tree. This clearly does not fit with what previous work has revealed about the dynamics of the epidemic, and it may reflect the limited power of these statistics compared to models that take the full phylogeny into account.

## Within-host HIV

Within a single host infected with HIV, we might expect that selection driven by neutralising antibodies would be the primary driver of asymmetry in the phylogenetic tree of the viral envelope, as rates of diversifying selection are significantly higher in HIV-1 *env* in individuals with robust neutralising antibody responses [32]. However, this is not the only cause of asymmetry in a phylogeny. We re-examined the HIV *env* sequence data of a patient who was previously shown to have a slow rate of immune escape from neutralising antibodies [32]. There were 134 full-length *env* sequences available, collected from 13 time points sampled over 1,098 days of follow up (Fig 6a). This phylogeny was found to be asymmetric with the standard Sackin's index (p <0.0001), and was also significant using the number of cherries (p = 0.028).

Correcting for the tip sampling with the permutation test, the number of cherries shows no evidence of global asymmetry in the phylogeny, though suggests there is some evidence of recent local asymmetry between 767 and 781 days from the present (Fig 6b). Sackin's index shows strong global asymmetry at the the root, which accumulates throughout the depth of the tree (Fig 6c). Within this, there are six individual nodes identified as having more asymmetric than expected subtrees below them (Fig 6d) with the Bonferroni correction, and 20 at the unadjusted 5% level. If we consider the q-value instead, there are 14 nodes with a q-value of 2.5% in the upper tail (for a 5% two-tailed test), suggesting that less than one of them (0.35) will be a false negative.

Examining the skyline plot for these data (<u>S1 Fig</u>) does not indicate any deviations from the null model. There are two distinct clades circulating within the patient at the same time in the tree, and if these clades were non-overlapping in time, we would see a pronounced dip in the





Fig 5. Asymmetry in the Sierra Leone ebola epidemic. a) Tree of 78 ebola virus whole genome sequences. b) Observed cumulative number of cherries over time (black), with results from permuted trees (grey). Inset histogram shows global results. Red lines show the medoid (solid) and 95% confidence interval of the permuted results (dashed). c) Trajectories for the cumulative Sackin's index. d) Node effect on Sackin's index over time.

skyline plot. This is not the case, with the effective population size instead showing steady exponential growth. This pattern and treeshape is reminiscent of the inter-subtype competition identified by Ferguson, Galvani and Bush [41].

## Discussion

In this paper, we have presented a framework to quantify asymmetry in phylogenetic trees where the tips have been sampled at different times. Previously, it has been highlighted that understanding the link between a tree topology and the evolutionary processes that gave rise to it is difficult [6, 42], which is further confounded by the fact that standard tests for asymmetry are implicitly biased in trees with heterochronous sampling [5]. The permutation test described



Fig 6. Within-host asymmetry is not always due to immune selection. a) Tree of 134 HIV envelope sequences from patient 83 [32]. b) Observed cumulative number of cherries over time (black), with results from permuted trees (grey). Inset histogram shows global results. Red lines show the medoid (solid) and 95% confidence interval of the permuted results (dashed). c) Trajectories for the cumulative Sackin's index. d) Node effect on Sackin's index over time. Nodes identified as significantly more asymmetric than expected with the Bonferroni correction are marked with a filled red circle in a), and those which are significant at an unadjusted p-value of 5% are shown by an open red circle.

here allows an observed phylogenetic tree to be compared to a distribution of coalescence trees, conditional on the same internal node and tip sampling structure. This is in contrast to the Temporal Clustering (TC) statistic proposed by Gray et al. [43], which tests for a 'temporal signal' in a tip-dated phylogeny, whereby sequences sampled around the same time are found clustered together in the tree and among these is the ancestor of any clade with sampling dates closer to the present. Their statistic permutes the tips with a fixed tree, whereas the test presented here permutes the tree conditional on the observed temporal structure in the form of tip sampling dates and internal node times. Trees with high temporal clustering have a higher

potential for false positives from the standard global tests. The three datasets presented in this paper all have a strong temporal signal according to their TC statistic. However, when we control for their temporal structures, they display different levels of asymmetry.

Although only Sackin's index and the number of cherries were illustrated here, the permutation test can be extended to other metrics of asymmetry including Colless' index [10], Shao and Sokal's balance statistics  $B_1$  and  $B_2$  [44], and the shape statistics of Agapow and Purvis [45, 46]. These statistics use varying measures of topological distance to quantify asymmetry, meaning they tend to be biased when tips are sampled earlier in the tree and have fewer nodes connecting them to the root and other tips of the tree. Given that the number of cherries and Sackin's index did not have the power to identify the asymmetry present in the ebola tree, it may well be worth considering a wider range of statistics alongside the permutation test if there is strong external suggestion of asymmetry in the tree. It is important to note that the branch lengths in a phylogeny can also convey important information about the dynamics of disease. The kernel function of Poon et al. [42] accounts for differences in branch lengths when comparing multiple trees, but cannot be used to statistically assess a single observed tree on its own. However, our permutation test could be used alongside this method to calculate the distance between the observed tree and simulated null trees. Additionally, the topology and the branch lengths of a viral phylogeny are not necessarily equivalent to the underlying transmission tree [47], and therefore it is important to be aware of the possible discrepancy in equating asymmetry in the phylogeny with asymmetry in transmission.

Generally, more complicated models will better fit the data. However, increased model complexity can be computationally intensive. As such, the model that is considered the best comes from a balance of the scientific relevance (the biological plausibility), the goodness of fit, and complexity [48]. While this usually relies on some simplifying assumptions, these are often violated—such as the assumption of a randomly mixing population. As a result, it is important to bear in mind the overall fit of the model to data. In the Bayesian framework, posterior predictive simulation is widely used for model checking, but despite recommendations for its use in the literature [49–53], it remains underutilised in the field of phylogenetics. In addition, these tests are often only possible alongside or once the analysis has been completed, after much computational effort. Since the base topology can often be recovered relatively quickly and accurately, our permutation test represents a quick method for checking whether the assumption of random mixing is supported, or whether there is evidence of asymmetry and therefore heterogeneity in the population.

We simply test for evidence of asymmetry in an observed tree, which can arise in the tree due to many processes in the underlying population such as contact rates and population structure [5, 7]. As evidenced with the within-host HIV data, it is not necessarily simple to interpret the underlying cause of local asymmetry being detected. It might be preferable to control for certain aspects of asymmetry occurring in the tree (that is, allow for some specific asymmetry in the null model), and see if there is significant evidence for further imbalance beyond that expected under the null model. However, methods that have become standard for inferring structure in the phylogenetic tree, such as the phylogeographic approach of Lemey et al. [54], make the assumption that the tree branching structure is not affected by the heterogeneity in the population (i.e. the population is randomly mixing, and the discrete trait model is simply overlaid over the tree). Thus, our permutation test can be used to justify whether this is an appropriate assumption, or whether it might be more advisable to use a more complex model such as the structured coalescent [55, 56].

Our approach is fast, has a free software implementation, and can offer important additional insights by highlighting potential lack of goodness-of-fit of commonly used coalescent and birth-death models.

# **Supporting Information**

**S1 Fig. Skyride ride plot for the within-host HIV phylogeny showing exponential growth.** Dashed lines show the 95% confidence interval, and red vertical lines indicate the timing of nodes evidence of higher than expected asymmetry in the tree. (TIFF)

**S1 Table.** Coverage probabilities for the permutation tests for the cumulative Sackin's index (a), node effect on Sackin's index (b) and cumulative number of cherries (c). Within datasets, for each permuted tree from the null distribution the probability of seeing a statistic as or more asymmetric in the remaining 9,999 permuted trees was calculated. The number of significant nodes was recorded, with the coverage probability being the proportion of trees for which there were no nodes with evidence of asymmetry. For a two-tailed hypothesis test with 95% confidence, we would expect the coverage to be around 97.5% showing that the unadjusted p-values give more false positives than expected, and that all adjustments should be conservative.

(PDF)

# Acknowledgments

Many thanks to Andrew Rambaut (University of Edinburgh) for providing the ebola phylogeny.

# **Author Contributions**

Conceived and designed the experiments: BLD SDWF. Performed the experiments: BLD SDWF. Analyzed the data: BLD SDWF. Contributed reagents/materials/analysis tools: BLD SDWF. Wrote the paper: BLD SDWF.

## References

- Pybus OG, Rambaut A (2009) Evolutionary analysis of the dynamics of viral infectious disease. Nat Rev Genet 10: 540–50. PMID: <u>19564871</u>
- Grenfell BT, Pybus OG, Gog JR, Wood JLN, Daly JM, et al. (2004) Unifying the epidemiological and evolutionary dynamics of pathogens. Science 303: 327–32. doi: <u>10.1126/science.1090727</u> PMID: <u>14726583</u>
- Frost SDW, Volz EM (2010) Viral phylodynamics and the search for an 'effective number of infections'. Phil Trans R Soc B 365: 1879–90. doi: 10.1098/rstb.2010.0060 PMID: 20478883
- 4. Dearlove B, Wilson DJ (2013) Coalescent inference for infectious disease: meta-analysis of hepatitis C. Phil Trans R Soc B 368: 20120314. doi: 10.1098/rstb.2012.0314 PMID: 23382432
- Frost SDW, Volz EM (2013) Modelling tree shape and structure in viral phylodynamics. Phil Trans R Soc B 368: 20120208. doi: 10.1098/rstb.2012.0208 PMID: 23382430
- Mooers AO, Heard SB (1997) Inferring evolutionary process from phylogenetic tree shape. Q Rev Biol 72: 31–54. doi: 10.1086/419657
- Leventhal GE, Kouyos R, Stadler T, Wyl VV, Yerly S, et al. (2012) Inferring epidemic contact structure from phylogenetic trees. PLoS Comput Biol 8: e1002413. doi: <u>10.1371/journal.pcbi.1002413</u> PMID: <u>22412361</u>
- 8. Pompei S, Loreto V, Tria F (2012) Phylogenetic properties of RNA viruses. PLoS One 7: e44849. doi: 10.1371/journal.pone.0044849 PMID: 23028645
- 9. Blum MGB, François O (2005) On statistical tests of phylogenetic tree imbalance: the Sackin and other indices revisited. Math Biosci 195: 141–53. doi: 10.1016/j.mbs.2005.03.003 PMID: 15893336
- Colless DH (1982) Phylogenetics: the theory and practice of phylogenetic systematics. Syst Zool 31: 100–104. doi: 10.2307/2413420
- 11. Sackin MJ (1972) "Good" and "bad" phenograms. Syst Zool 21: 225–226. doi: 10.2307/2412292

- McKenzie A, Steel M (2000) Distributions of cherries for two models of trees. Math Biosci 164: 81–92. doi: <u>10.1016/S0025-5564(99)00060-7</u> PMID: <u>10704639</u>
- Kirkpatrick M, Slatkin M (1993) Searching for evolutionary patterns in the shape of a phylogenetic tree. Evolution 47: 1171–1181. doi: <u>10.2307/2409983</u>
- Bortolussi N, Durand E, Blum M, François O (2006) apTreeshape: statistical analysis of phylogenetic tree shape. Bioinformatics 22: 363–364. doi: <u>10.1093/bioinformatics/bti798</u> PMID: <u>16322049</u>
- 15. R Core Team (2013). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL <u>http://www.r-project.org/</u>.
- Rogers JS (1996) Central moments and probability distributions of three measures of phylogenetic tree imbalance. Syst Biol 45: 99–110. doi: <u>10.1093/sysbio/45.1.99</u>
- Blum MGB, François O, Janson S (2006) The mean, variance and limiting distribution of two statistics sensitive to phylogenetic tree balance. Ann Appl Probab 16: 2195–2214. doi: <u>10.1214/</u> 105051606000000547
- Maechler M, Rousseeuw P, Struyf A, Hubert M, Hornik K (2014). cluster: Cluster analysis basics and extensions. R package version 1.15.2.
- 19. Ernst MD (2004) Permutation methods: a basis for exact inference. Stat Sci 19: 676–685. doi: 10.1214/ 08834230400000396
- 20. Holm S (1979) A simple sequentially rejective multiple test procedure. Scand J Stat 6: 65–70.
- **21.** Hochberg Y (1988) A sharper Bonferroni procedure for multiple tests of significance. Biometrika 75: 800–802. doi: <u>10.1093/biomet/75.4.800</u>
- Hommel G (1988) A stagewise rejective multiple text procedure based on a modified Bonferroni test. Biometrika 75: 383–6. doi: <u>10.1093/biomet/75.2.383</u>
- Benjamini Y, Hochberg Y (1995) Controlling the false discovery rate: a practical and powerful approach to multiple testing. J R Stat Soc B 57: 289–300.
- 24. Benjamini Y, Yekutieli D (2001) The control of the false discovery rate in multiple testing under depencency. Ann Stat 29: 1165–1188.
- Storey JD (2002) A direct approach to false discovery rates. J R Stat Soc B 64: 479–498. doi: <u>10.1111/</u> <u>1467-9868.00346</u>
- Storey JD, Tibshirani R (2003) Statistical significance for genomewide studies. Proc Natl Acad Sci USA 100: 9440–9445. doi: <u>10.1073/pnas.1530509100</u> PMID: <u>12883005</u>
- Storey JD (2003) The positive false discovery rate: a Bayesian interpretation and the q-value. Ann Stat 31: 2013–2035. doi: 10.1214/aos/1074290335
- Anderson CNK, Ramakrishnan U, Chan YL, Hadly EA (2005) Serial SimCoal: a population genetics model for data from multiple populations and points in time. Bioinformatics 21: 1733–4. doi: <u>10.1093/</u> <u>bioinformatics/bti154</u> PMID: <u>15564305</u>
- Drummond AJ, Suchard MA, Xie D, Rambaut A (2012) Bayesian phylogenetics with BEAUti and the BEAST 1.7. Mol Biol Evol 29: 1969–73. doi: <u>10.1093/molbev/mss075</u> PMID: <u>22367748</u>
- Gire SK, Goba A, Andersen KG, Sealfon RSG, Park DJ, et al. (2014) Genomic surveillance elucidates Ebola virus origin and transmission during the 2014 outbreak. Science 345: 1369–1372. doi: <u>10.1126/science.1259657</u> PMID: <u>25214632</u>
- Wallace RG, HoDac H, Lathrop RH, Fitch WM (2007) A statistical phylogeography of influenza A H5N1. Proc Natl Acad Sci USA 104: 4473–4478. doi: 10.1073/pnas.0700435104 PMID: 17360548
- Frost SDW, Wrin T, Smith DM, Kosakovsky Pond SL, Liu Y, et al. (2005) Neutralizing antibody responses drive the evolution of human immunodeficiency virus type 1 envelope during recent HIV infection. Proc Natl Acad Sci USA 102: 18514–9. doi: <u>10.1073/pnas.0504658102</u> PMID: <u>16339909</u>
- Edgar RC (2004) MUSCLE: multiple sequence alignment with high accuracy and high throughput. Nucleic Acids Res 32: 1792–7. doi: <u>10.1093/nar/gkh340</u> PMID: <u>15034147</u>
- Tavaré S (1986) Some probabilistc and statistical problems in the analysis of DNA sequences. Lectures Math Life Sci 17: 57–86.
- Drummond AJ, Ho SYW, Phillips MJ, Rambaut A (2006) Relaxed phylogenetics and dating with confidence. PLoS Biol 4: e88. doi: <u>10.1371/journal.pbio.0040088</u> PMID: <u>16683862</u>
- Palacios JA, Minin VN (2012) Integrated nested Laplace approximation for Bayesian nonparametric phylodynamics. In: Proceedings of the Twenty-Eighth Conference on Uncertainty in Artificial Intelligence (UAI-12), Corvallis, Oregon: AUAI Press. pp. 726–735.
- **37.** WHO (2015). Ebola data and statistics: Situation summary. Data published on 15 April 2015. URL http://apps.who.int/gho/data/view.ebola-sitrep.ebola-summary-20150415?lang = en.

- Volz E, Kosakovsky Pond S (2014) Phylodynamic analysis of Ebola virus in the 2014 Sierra Leone epidemic. PLoS Curr outbreaks. doi: 10.1371/currents.outbreaks.6f7025f1271821d4c815385b08f5f80e
- Stadler T, Kühnert D, Rasmussen DA, du Plessis L (2014) Insights into the early epidemic spread of Ebola in Sierra Leone provided by viral sequence data. PLoS Curr outbreaks. doi: <u>10.1371/currents.</u> <u>outbreaks.02bc6d927ecee7bbd33532ec8ba6a25f</u>
- **40.** Łuksza M, Bedford T, Lässig M (2014) Epidemiological and evolutionary analysis of the 2014 Ebola virus outbreak. ArXiv e-prints: <a href="http://arxiv.org/abs/1411.1722">http://arxiv.org/abs/1411.1722</a>.
- **41.** Ferguson NM, Galvani AP, Bush RM (2003) Ecological and immunological determinants of influenza evolution. Nature 422: 428–433. doi: <u>10.1038/nature01509</u> PMID: <u>12660783</u>
- Poon AFY, Walker LW, Murray H, McCloskey RM, Harrigan PR, et al. (2013) Mapping the shapes of phylogenetic trees from human and zoonotic RNA viruses. PLoS One 8: e78122. doi: <u>10.1371/journal.</u> pone.0078122 PMID: 24223766
- Gray RR, Pybus OG, Salemi M (2011) Measuring the Temporal Structure in Serially-Sampled Phylogenies. Methods Ecol Evol 2: 437–445. doi: 10.1111/j.2041-210X.2011.00102.x PMID: 22121470
- 44. Shao KT, Sokal RR (1990) Tree balance. Syst Zool 39: 266–276. doi: 10.2307/2992186
- 45. Agapow PM, Purvis A (2002) Power of eight tree shape statistics to detect nonrandom diversification: a comparison by simulation of two models of cladogenesis. Syst Biol 51: 866–872. doi: <u>10.1080/</u>10635150290102564 PMID: 12554452
- 46. Purvis A, Katzourakis A, Agapow PM (2002) Evaluating phylogenetic tree shape: two modifications to Fusco & Cronk's method. J Theor Biol 214: 99–103. doi: <u>10.1006/jtbi.2001.2443</u> PMID: <u>11786035</u>
- Romero-Severson E, Skar H, Bulla I, Albert J, Leitner T (2014) Timing and order of transmission events is not directly reflected in a pathogen phylogeny. Mol Biol Evol 31: 2472–2482. doi: <u>10.1093/molbev/</u> <u>msu179</u> PMID: <u>24874208</u>
- Steel M (2005) Should phylogenetic models be trying to 'fit an elephant'? Trends Genet 21: 307–9. doi: 10.1016/j.tig.2005.04.001 PMID: 15922824
- Huelsenbeck JP, Ronquist F, Nielsen R, Bollback JP (2001) Bayesian inference of phylogeny and its impact on evolutionary biology. Science 294: 2310–4. doi: 10.1126/science.1065889 PMID: 11743192
- Bollback JP (2002) Bayesian model adequacy and choice in phylogenetics. Mol Biol Evol 19: 1171– 80. doi: 10.1093/oxfordjournals.molbev.a004175 PMID: 12082136
- Bollback JP (2005) Posterior mapping and posterior predictive distributions. In: Nielsen R, editor, Statistical Methods in Molecular Evolution, New York: Springer. pp. 439–462.
- Brown JM, ElDabaje R (2009) PuMA: Bayesian analysis of partitioned (and unpartitioned) model adequacy. Bioinformatics 25: 537–8. doi: 10.1093/bioinformatics/btn651 PMID: 19098028
- Drummond AJ, Suchard MA (2008) Fully Bayesian tests of neutrality using genealogical summary statistics. BMC Genet 9: 68. doi: 10.1186/1471-2156-9-68 PMID: 18976476
- Lemey P, Rambaut A, Drummond AJ, Suchard MA (2009) Bayesian phylogeography finds its roots. PLoS Comput Biol 5: e1000520. doi: <u>10.1371/journal.pcbi.1000520</u> PMID: <u>19779555</u>
- 55. Hudson RR (1991) Gene genealogies and the coalescent process. Oxford Surv Evol Biol 7: 1–44.
- Notohara M (1990) The coalescent and the genealogical process in geographically structured population. J Math Biol 29: 59–75. doi: 10.1007/BF00173909 PMID: 2277236